Author's response to reviews

Title: Antibodies against Coxiella burnetii and pregnancy outcome during the 2007-2008 Q fever outbreaks in the Netherlands

Authors:

Wim van der Hoek (wim.van.der.hoek@rivm.nl)
Jamie C.E. Meekelenkamp (J.Meekelenkamp@jbz.nl)
Alexander C.A.P. Leenders (A.Leenders@jbz.nl)
Nancy Wijers (n.wijers@iq.umcn.nl.com)
Daan W. Notermans (daan.notermans@rivm.nl)
Chantal W.P.M. Hukkelhoven (chukkelhoven@perinatreg.nl)

Version: 5 Date: 8 February 2011

Author's response to reviews:

Style and order of the list of authors in the manuscript and in the submission system are now identical.

The changes have been made in the submission system. No changes were made in the manuscript.

Sorry for the inconvenience.